Author's response to reviews

Title: The influence of nativity and neighborhoods on breast cancer stage at diagnosis and survival among California Hispanic women

Authors:

Theresa HM Keegan (theresa.keegan@cpic.org)
Thu Quach (thu.quach@cpic.org)
Sarah Shema (sarah.shema@cpic.org)
Sally L Glaser (sally.glaser@cpic.org)
Scarlett L Gomez (scarlett.gomez@cpic.org)

Version: 7 Date: 2 November 2010

Author's response to reviews:

We have reviewed the accepted article that was formatted by the journal. The attached version contains corrections to a few small typographical errors. In addition, we have removed "formerly the Northern California Cancer Center" from the authors' affiliations and instead note the former name of the Institution in the Acknowledgements section. The attached version of the manuscript is acceptable for publication.